Author's response to reviews

Title: The growth of a culture of evidence-based obstetrics in South Africa: a qualitative case study

Authors:

Karen L Daniels Ms (karen.daniels@mrc.ac.za)
Simon A Lewin Dr (simon.lewin@nokc.no)
Practihc Policy Group (karen.daniels@mrc.ac.za)

Version: 4 Date: 15 March 2011

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Footnotes: We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that you incorporate the footnotes into the text of the manuscript. Alternatively, you may wish to make an appendix section containing your footnotes.

- The footnote has been incorporated into the text.

Author list: the list of authors in the manuscript and the list of authors in the submission system should be identical, both in style and order. The preferred style is: First name Initial Last name (e.g. Joe F Bloggs)

- We have corrected this and have removed the middle initial for each of the authors in the submission system as per the text.

Study Groups: Members of a group should be listed in the acknowledgements (if anywhere). You have currently mentioned them in the authors' contributions. You can mention the contributions of the study group in the authors' contributions section but name individual members in the acknowledgments.

- This has been corrected according to the editor's specifications.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

- This has been corrected according to the editor's specifications.

References: please provide a functioning link to the websites in reference 42, 45

- We have removed reference 42 as we cannot provide a functioning link and we feel that the processes referred to are sufficiently cited.

- We have updated reference 45 (now 44). Although the conference database is no longer available online readers can obtain it from the conference organisers by using the link provided.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the submissions system

- This has been removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

- We are happy that there are no typographical errors.